Nurse staffing levels, missed vital signs and mortality in hospitals: retrospective longitudinal observational study

Peter Griffiths,^{1,2*} Jane Ball,^{1,2} Karen Bloor,³ Dankmar Böhning,⁴ Jim Briggs,⁵ Chiara Dall'Ora,^{1,2} Anya De Iongh,⁶ Jeremy Jones,¹ Caroline Kovacs,⁵ Antonello Maruotti,¹ Paul Meredith,^{2,7} David Prytherch,^{2,5,7} Alejandra Recio Saucedo,^{1,2} Oliver Redfern,⁵ Paul Schmidt,^{2,7} Nicola Sinden⁷ and Gary Smith⁸

¹Health Sciences, University of Southampton, Southampton, UK
²National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care Wessex, Southampton, UK
³Health Sciences, University of York, York, UK
⁴Mathematical Sciences, University of Southampton, Southampton, UK
⁵Centre for Healthcare Modelling and Informatics, University of Portsmouth, Portsmouth UK
⁶Independent lay researcher c/o National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care, Southampton, UK
⁷Clinical Outcomes Research Group, Portsmouth Hospitals NHS Trust, Queen Alexandra Hospital, Portsmouth, UK

⁸Health and Social Sciences, Bournemouth University, Bournemouth, UK

*Corresponding author Peter.griffiths@soton.ac.uk

Declared competing interests of authors: All authors received grant support from the National Institute for Health Research (NIHR) for the conduct of this study, remuneration for employment from their employing institution and may be in receipt of other funding for non-commercial independent research outside the current work. Peter Griffiths is a member of the advisory group for NHS Improvement's ongoing work on safe staffing in adult inpatient wards (unpaid). Chiara Dall'Ora received grants from the NIHR Collaboration for Leadership in Applied Health Research and Care (CLAHRC) Wessex to support her PhD. Anya De longh received personal fees for freelance work and employment outside the submitted work from Help & Care (Bournemouth, UK), Dorset HealthCare University NHS Foundation Trust, Health Education England/Skills for Health, University of Southampton, University of Cambridge, King's College London, Queen Mary University London and the NIHR. Gary Smith receives non-financial support from non-commercial organisations: Resuscitation Council UK, Royal College of Physicians of London, International Society of Rapid Response Systems and the NIHR outside the submitted work. For some of the period of the research, Portsmouth Hospitals NHS Trust (PHT), which employed Paul Meredith, David Prytherch, Gary Smith, Nicola Sinden and Paul Schmidt, had a royalty agreement with The Learning Clinic Limited (now part of System C Healthcare Ltd, Maidstone, Kent) to pay for the use of PHT intellectual property within The Learning Clinic Limited's

Vitalpac system, which was used to collect the vital signs observations data used in the project. Until October 2015, the wives of David Prytherch and Gary Smith held minority shareholdings in The Learning Clinic Limited. Paul Schmidt held a personal shareholding prior to the commencement of the study.

Published November 2018 DOI: 10.3310/hsdr06380

Plain English summary

Nurse staffing levels, missed vital signs and mortality in hospitals Health Services and Delivery Research 2018; Vol. 6: No. 38 DOI: 10.3310/hsdr06380

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Shortages of nurses have been linked to poor care in hospitals but it is hard to specify the number and type of staff needed, or to be sure that staff levels are really the cause of the problem. Higher staffing levels cost more, so there are financial perspectives to consider. The aim of our study was to see if observations of 'vital signs' (such as pulse and blood pressure) were affected by low staffing levels, and whether or not this might explain higher death rates when staffing is low. If this is the case, nurses' ability to complete the necessary observations might be used to see how safe staffing levels are.

We used hospital records about nurses on duty, patient records and patients' observations from 32 wards over 3 years. We used statistical models to see if patients were more likely to die, have a longer stay and have observations missed if they were on wards where there were fewer nurses than if they were on wards with more nurses. We used the results to estimate what might happen if staffing levels were changed and what the costs would be.

We found that patients who spent time on wards with fewer than the usual number of fully qualified nurses were more likely to die, or to stay in hospital for longer. When staffing was lower, more observations were missed, and this was related to higher death rates, but low staffing could not explain why most observations were being missed. Levels of health-care assistants, who provide support to nurses, were also important, but deaths could be reduced at lowest cost by replacing some assistants with qualified nurses, although our design means that we cannot be certain of cause and effect.

Although missed observations explain some of the links between nurse staffing levels and hospital death rates, these records cannot easily show that staffing levels are safe. Assistants are unlikely to make up for a shortfall of qualified nurses.

© Queen's Printer and Controller of HMSO 2018. This work was produced by Griffiths et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Health Services and Delivery Research

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme or programmes which preceded the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HS&DR programme

The Health Services and Delivery Research (HS&DR) programme, part of the National Institute for Health Research (NIHR), was established to fund a broad range of research. It combines the strengths and contributions of two previous NIHR research programmes: the Health Services Research (HSR) programme and the Service Delivery and Organisation (SDO) programme, which were merged in January 2012.

The HS&DR programme aims to produce rigorous and relevant evidence on the quality, access and organisation of health services including costs and outcomes, as well as research on implementation. The programme will enhance the strategic focus on research that matters to the NHS and is keen to support ambitious evaluative research to improve health services.

For more information about the HS&DR programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hsdr

This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 13/114/17. The contractual start date was in June 2015. The final report began editorial review in September 2017 and was accepted for publication in February 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2018. This work was produced by Griffiths *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk